Author's response to reviews

Title: Diffusion Tensor Imaging Correlates with Lesion Volume in Cerebral Hemisphere Infarctions

Authors:

Maija E Rossi (maija.rossi@pshp.fi)
Eeva Jason (eeva.jason@uta.fi)
Silvia Marchesotti (sissymarche@gmail.com)
Prasun Dastidar (prasun.dastidar@pshp.fi)
Jyrki Ollikainen (jyrki.ollikainen@pshp.fi)
Seppo Soimakallio (seppo.soimakallio@pshp.fi)

Version: 4 Date: 8 September 2010

Author's response to reviews:

Dear scientific editor Mrs. Chap and dear reviewers,

Thank you for having re-evaluated our manuscript and accepted our revisions. We have completed the three remaining requests within the manuscript.

First, we have inserted the missing second power in the equation for radial diffusivity on page 14.

Second, we now indicate the missing fiber trackts with arrows in Figure 4, and we also explain this in the figure legend.

Third, we have once more inspected the SD in Table 2.

Finally, we have made slight modifications in the style of the title page to conform to the journal style.

We hope to have answered your requests but we are happy to discuss any details you still may find relevant to this study.

Best regards,

Maija Rossi